Author’s response to reviews

Title: MEOPAeDent trial protocol - observational study of Equimolar Mixture of Oxygen and Nitrous Oxide (EMONO) effects in pediatric dentistry

Authors:

Tony PRUD'HOMME (tony.prudhomme@univ-nantes.fr)
Sylvie DAJEAN-TRUTAUD (sylvie.dajean-trautaud@univ-nantes.fr)
Morgane ROUSSELET (morgane.rousselet@chu-nantes.fr)
Fanny FEUILLET (fanny.feuillet@chu-nantes.fr)
Marjorie CARPENTIER-CHERAUD (marjorie.CARPENTIERCHERAUD@chu-nantes.fr)
Olivier BONNOT (olivier.bonnot@chu-nantes.fr)
Isabelle HYON (Isabelle.HYON@chu-nantes.fr)
Marie GRALL-BRONNEC (marie.bronnec@chu-nantes.fr)
Serena LOPEZ-CAZAUX (serena.lopez-cazaux@univ-nantes.fr)
Caroline VICTORRI-VIGNEAU (caroline.vigneau@chu-nantes.fr)

Version: 1 Date: 12 Nov 2018

Author’s response to reviews:

Editor Comments:

1) Please change the headings “Methods/design” to “Methods” in the abstract and the main text.

Change made (page 2 and 4)

2) Please change the date of the trial registration from “2 Mars 2018” to “2 March 2018” and, in addition, remove the Unique Protocole ID.

Change made (page 2)

3) Under the heading “Ethics approval and consent to participate”, please remove the information about registering on clinicaltrials.gov.
4) Unfortunately, there are some minor language issues throughout the manuscript. Please ensure that you have thoroughly checked your manuscript for any other language errors. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

Review by Springer Nature (with Certificate)